Author’s response to reviews

Title: Genetic variants in FBLIM1 gene do not contribute to SAPHO syndrome and chronic recurrent multifocal osteomyelitis in typical patient groups

Authors:
Gunter Aßmann (gunter.assmann@uks.eu)
Michaela Köhm (Michaela.Koehm@ime.fraunhofer.de)
Volker Schuster (Volker.Schuster@medizin.uni-leipzig.de)
Frank Behrens (Frank.Behrens@ime.fraunhofer.de)
Rotraut Mössner (rmoessn@gwdg.de)
Nina Magnolo (Nina.Magnolo@ukmuenster.de)
Vinzenz Oji (Vinzenz.Oji@ukmuenster.de)
Harald Burkhardt (Harald.Burkhardt@kgu.de)
Ulrike Hüffmeier (Ulrike.Hueffmeier@uk-erlangen.de)

Version: 4 Date: 23 Apr 2020

Author’s response to reviews:

Dear Editor, we changed the manuscript by including the final editorial requests. We hope that the manuscript is now acceptable and look forward to hearing from you. Best regards Ulrike Hüffmeier on behalf of all coauthors